Author's response to reviews

Title: Essential health information available for India in the public domain on the Internet

Authors:

Magdalena Z Raban (mrab2647@mail.usyd.edu.au)
Rakhi Dandona (rdandona@med.usyd.edu.au)
Lalit Dandona (dandona@u.washington.edu)

Version: 6 Date: 15 June 2009

Author's response to reviews: see over
15 June 2009

Editor

*BMC Public Health*

Dear Editor,

RE: Manuscript 4852768262299956

Thank you for inviting further minor revisions of our manuscript. As suggested, the title of the manuscript has now been changed to clearly state that this review was based on a search of the World Wide Web: “Essential health information available for India in the public domain on the Internet”. Similarly, this is now explicitly stated in all sections of the manuscript.

We thank BMC Public Health for considering our manuscript for publication.

Yours sincerely,

Magdalena Raban
Rakhi Dandona
Lalit Dandona